Author's response to reviews

Title: Measurement of a model of dissemination for health care: Toward a testable theory

Authors:

Joan M Cook (Joan.Cook@Yale.edu)
Casey O'Donnell (O'donnell.Casey@gmail.com)
Stephanie Dinnen (Stephanie.Dinnen@yale.edu)
James C Coyne (JCoyne@mail.med.upenn.edu)
Josef I Ruzek (Josef.Ruzek@va.gov)
Paula P Schnurr (Paual.P.Schnurr@dartmouth.edu)

Version: 2 Date: 19 December 2011

Author's response to reviews:

We have addressed the structure of the Abstract to match those listed in the Guidelines to Authors. Additionally, we have structured the body of the paper around the subject headings background, methods, results, and discussion. As this manuscript details the rationale and development of a quantitative survey and semi-structured interview guide operationalizing a testable model of dissemination of innovations in healthcare settings it details/involves no human subjects research and thus this section has not been included.

Further, we have added a statement of no-competing interests and a statement of authors' contributions at the end of the paper but before the references.